Author's response to reviews

Title: Percutaneous septal ablation for left mid-ventricular obstructive hypertrophic cardiomyopathy: a case report

Authors:

Istemihan Tengiz (dritengiz@yahoo.com)
Ertugrul Ercan (ertugrulercan@yahoo.com)
Emin Alioglu (emilaliyev@mynet.com)
Ugur O. Turk (droturk@yahoo.com)

Version: 8 Date: 23 December 2005

Author's response to reviews:

Dear editor,
Abstract section structured into Background, Case presentation, Conclusions and Authors' contributions provided a brief description of each of the author's participation in the study.
Best regards,
Istemihan Tengiz MD, FESC